



## Journal of Applied Accounting Research

The incremental information content of audit opinion Panayiotis Tahinakis Michalis Samarinas

## Article information:

To cite this document: Panayiotis Tahinakis Michalis Samarinas , (2016),"The incremental information content of audit opinion", Journal of Applied Accounting Research, Vol. 17 Iss 2 pp. 139 - 169 Permanent link to this document: http://dx.doi.org/10.1108/JAAR-01-2013-0011

Downloaded on: 18 July 2016, At: 10:43 (PT) References: this document contains references to 113 other documents. To copy this document: permissions@emeraldinsight.com The fulltext of this document has been downloaded 159 times since 2016\*

## Users who downloaded this article also downloaded:

(2016),"Audit quality and earnings management: evidence from Jordan", Journal of Applied Accounting Research, Vol. 17 Iss 2 pp. 170-189 http://dx.doi.org/10.1108/JAAR-09-2014-0089

(2016), "Discussion of "articulating accounting principles: Classical accounting theory as the pursuit of 'explanation by embodiment'", Journal of Applied Accounting Research, Vol. 17 Iss 2 pp. 136-138 http://dx.doi.org/10.1108/JAAR-03-2016-0028

(2016),"Articulating accounting principles: Classical accounting theory as the pursuit of "explanation by embodiment", Journal of Applied Accounting Research, Vol. 17 Iss 2 pp. 118-135 http://dx.doi.org/10.1108/JAAR-01-2014-0017

Access to this document was granted through an Emerald subscription provided by emeraldsrm:423451 []

## **For Authors**

If you would like to write for this, or any other Emerald publication, then please use our Emerald for Authors service information about how to choose which publication to write for and submission guidelines are available for all. Please visit www.emeraldinsight.com/authors for more information.

### About Emerald www.emeraldinsight.com

Emerald is a global publisher linking research and practice to the benefit of society. The company manages a portfolio of more than 290 journals and over 2,350 books and book series volumes, as well as providing an extensive range of online products and additional customer resources and services.

Emerald is both COUNTER 4 and TRANSFER compliant. The organization is a partner of the Committee on Publication Ethics (COPE) and also works with Portico and the LOCKSS initiative for digital archive preservation.

\*Related content and download information correct at time of download.

# The incremental information content of audit opinion

Panayiotis Tahinakis and Michalis Samarinas Department of Accounting and Finance, University of Macedonia, Thessaloniki, Greece

#### Abstract

**Purpose** – The purpose of this paper is to examine the incremental information content of audit opinion while considering opinion determinants, such as auditor and auditee size, or a firm's financial state. **Design/methodology/approach** – A market valuation model is employed using US firm data collected over 30 years. The model relates stock returns to earnings and incorporates as additional variables auditors' opinion types, opinion determinants and their interactions with audit expression. **Findings** – The findings suggest that audit opinion has a significant market impact. The estimated positive or negative information content of the audit opinion types is associated with certain opinion determinants, such as auditor and auditee size and a firm's financial state.

**Research limitations/implications** – Additional firm-year observations regarding certain opinion qualifications could benefit future research.

**Practical implications** – This study offers useful insights by demonstrating the importance of auditing profession to the users of financial statements. It examines investors' perception of each audit opinion type and the conditions under which this expression has the most serious effects. The results demonstrate the role of audit opinion and its cause-effect relationship with various economic events, allowing regulators not only to track the efficiency of various audit policy changes but also act preventively and amend the regulatory framework.

**Originality/value** – This paper empirically supports the significance of the auditing process and audit opinions by examining investor perceptions. It employs a value relevance model, in contrast to market-based research that adopts an event study methodology.

Keywords Stock returns, Audit opinion, Auditor's report, Market reaction

Paper type Research paper

#### 1. Introduction

In theory, the auditing process ensures trust, a key factor necessary in the operation of financial markets. It provides legitimacy (Power, 2003) and assures investors, analysts and market competitors that information extracted from financial statements is valid. However, there is some doubt in literature regarding this claim. Rajgopal and Venkatachalam (2011), Biddle *et al.* (2009) and Lev and Zarowin (1999) discuss a systematic decline in the usefulness of financial information for investors. This phenomenon, created by the deterioration in financial reporting quality, is interdependent with the quality of the auditing process (Hope *et al.*, 2013; Walker, 2013), raising questions about the validity that the market attributes to audit expression (Aziz and Omoteso, 2014). Furthermore, a series of corporate collapses shined a spotlight, and thus spurred discussions, on auditor independence (Adelopo, 2009). In this context, this paper attempts to contribute to this discussion by examining whether audit opinion adds credibility and value to financial statements information



Journal of Applied Accounting Research Vol. 17 No. 2, 2016 pp. 139-169 © Emerald Group Publishing Limited 09675426 DOI 10.1108/JAAR-01-2013-0011

Incremental information content

139

The authors would like to thank the Editor as well as the two anonymous referees for their help and valuable remarks that made the completion of this paper possible. They would also like to thank Professor Negakis CI, Associate Professor Kousenidis DV and the Lecturer Ladas AC for their great help and comments during the preparation of this paper.

provided to the investing public. By examining US firm data collected for a 30-year period, the empirical analysis explores the differential impact that each audit opinion type has on investor decisions while accounting for audit opinion determinants that simultaneously affect them.

Previous studies have produced sparse and mixed results (Ittonen, 2012). They vividly demonstrated the created antithesis presenting a wide range of results that vary from a positive/negative reaction to the announcement of an auditor's opinion, to a complete lack of it. This illustrates the difficulty in addressing the issue.

This study presents an alternative methodological rationale by employing a variant of the Easton and Harris (1991) model, augmented by a set of intercept and slope dummy variables. The empirical evidence of the study suggests that both unqualified and qualified audit opinions have a significant market impact. These findings support the argument that the positive or negative information content derived from audit opinion types is associated with certain audit opinion determinants, such as auditor and auditee size and a firm's financial well-being. Furthermore, the results show that the findings are consistent, implying that despite the shocks created by various economic events, the value relevance of audit opinion persists.

This paper contributes to auditing literature by focusing on the investing public's validation of audit opinion credibility, first, by studying different patterns in auditing and investment behaviour observed over 30 years. This study implies that both shareholders and potential investors adjust their investment decisions based on the value they attribute to information that accompanies and validates disclosed financial statements. Furthermore the current research results will provide auditors with a useful tool to assess not only the audit opinion types that have significant market impacts but also the conditions under which each opinion has the most serious effect. Finally, the findings demonstrate the role of audit opinion and its cause-effect relationship with various economic events (e.g. stock market crashes, financial crises and expansion periods), allowing regulators to act preventively, amend the regulatory framework when needed and track the efficiency of various audit policy changes.

The remainder of the paper is organised as follows: Section 2 provides a literature review while Section 3 elaborates on methodological issues and focuses on hypothesis development and model formation. Section 4 presents the data employed in the study and Section 5 gives the empirical analysis results. Finally, Section 6 summarises the paper and offers implications for further research.

#### 2. Review of previous research

Prior research examined auditor expression and its effects on investors and highlighted an antithesis created by the contradiction of findings (Ittonen, 2012). Holt and Moizer (1990) attempted a research categorisation, adopted in this study, to outline the research to date (Figure 1).

In this classification, studies are categorised into two main groups: reaction and interpretation research. The "reaction studies" group which examines reader's reaction to the announcement of audit opinion, is further divided into two distinct subgroups – market and experiment-based studies.

Market-based studies examine abnormal returns, potentially created by an auditor's report, employing an event study methodology (Ittonen, 2012). Their results seem contradictory. Some conclude that audit report's announcement has an insignificant effect or no effect at all on stock returns and prices (Al-Thuneibat *et al.*, 2008; Ameen *et al.*, 1994; Baskin 1972; Brito and Peres, 2006; Czernkowski *et al.*, 2010; Davis

1982; Dodd *et al.*, 1984; Firth 1978; Martinez *et al.*, 2004; Mittelstaedt *et al.*, 1992; Pucheta Martínez *et al.*, 2004). Specifically, Ameen *et al.* (1994) dismiss audit reports information emphasising the market's awareness of a problem.

Others have provided evidence that only certain types of qualifications affect investor decisions (Ball *et al.*, 1979; Banks and Kinney, 1982; Citron and Taffler, 1992; Chow and Rice, 1982; Dopuch *et al.*, 1986; Elliott, 1982; Fargher and Wilkins, 1998; Herbohn *et al.*, 2007; Soltani, 2000). Chen *et al.* (2000) examined modified audit opinions showing a negative market reaction. Numata and Takeda (2010) investigated the relationship between audit quality and market response providing evidence of negative stock market reactions to poor quality audit announcements.

The second subgroup – experiment-based studies – employs a questionnairebased methodological rationale. Test subjects like market or bank analysts, are given hypothetical financial statements along with various types of audit reports and are then asked to make a decision regarding, for example, a loan provision to the subject firm after examining the data provided. This research subcategory, also depicts contradictory findings and demonstrates the above mentioned antithesis. Estes and Reimer (1977), Houghton (1983), Robertson (1988) and Lin *et al.* (2003) argue that the audit report seems to bare no significant influence on the analysts. On the other hand, Firth (1980), Gul (1987), Bessell *et al.* (2003) and Guillamon (2003) stated that an auditor's report is a useful source of information for market participants and that certain types of qualifications affect share prices and financial decisions.

Interpretation studies – the second main group – take for granted that audit reports are important concentrating on their meaning (Holt and Moizer, 1990). This questionnaire-based approach considers how different wordings, meanings and readers create perceptions of the auditor's report importance (Bailey, 1981; Bailey *et al.*, 1983; Craswell, 1985; Dillard and Jensen, 1983; Holt and Moizer, 1990; Hatherly *et al.*, 1991; Innes *et al.*, 1997; Libby, 1979; Miller *et al.*, 1993; Slack *et al.*, 2010).

The qualitative nature of audit expression challenges the attempt to reach a robust conclusion (Lin *et al.*, 2003). Acknowledging these problems, this study focuses on how the market validates audit opinion effects on investors. However, event study methodology the most utilised research approach for the topic (Ittonen, 2012), stumbles upon a noticeable problem: determination of the exact event date. The announcement date of auditor's report is the event date employed for such analysis (Dopuch *et al.*, 1986; Ameen *et al.*, 1994). However, the problem with this date, as discussed by Elliott (1982) and Dopuch *et al.* (1986), is that once the auditor's opinion becomes public it can reach investors on several different occasions.

The date of the report's issuance coincides with that of financial statements, thus creating "noise" and a subsequent difficulty in separating the various effects.



Figure 1. Previous research classification

Statement on Auditing Standards (SAS) 103 allows identification of the problem. Its provisions stipulate the dating of a report as, either the date when sufficient evidence is obtained to support the auditor's opinion[1] or the date the auditor grants permission to use the report in connection with the financial statements[2]. Thus, the issue of an actual event date is argued to be a grey area.

#### 3. Methodology and hypothesis development

#### <sup>•</sup> 3.1 The relationship between stock returns, earnings and earnings changes

This study's methodological approach employs a value relevance model as a baseline for the estimation procedure. This model is selected for its ability to assess whether accounting and auditing information is used by investors in their valuation of a firm's equity (Barth *et al.*, 2001). It provides useful insights into how audit information impacts investment behaviour, and allows an examination of how specific factors determining audit opinion can affect primary users of financial statements (Concepts Statements No. 8; FASB, 2010) in their market decisions.

To conceptualise the relationship between the market and audit opinion with a value relevance model, several studies were considered that describe the relationship between stock returns or stock prices with various accounting measures (Easton and Harris, 1991; Easton *et al.*, 1992; Feltham and Ohlson, 1995; Harris and Ohlson, 1987; Ohlson, 1995). The choice of a returns model, more specifically, Easton and Harris' (1991) methodology is based on its characteristic of having fewer econometric limitations and its capacity to formally include uncertainty (Landsman and Magliolo, 1988; Kothari and Zimmerman, 1995). Drawing on Chen and Zhang's (2007) attempt to expand Easton and Harris' (1991) model by using other accounting information as explanatory variables for returns, this study employs a variant of Easton and Harris (1991) methodology: it relates stock returns with earnings and earnings changes but also includes audit opinion types and a series of control variables, thus extending the initial model. Easton and Harris' methodology is captured by the following mathematical representation:

$$R_{i,t} = a_0 + a_1 EARN_{i,t} + a_2 \Delta EARN_{i,t} + e_{i,t} \tag{1}$$

 $R_{i,t}$  represents stock returns that are estimated by calculating the annual change in the stock price, including dividends by ex-date and paid per share for the current fiscal year, divided by the stock price nine months prior to the fiscal year end. The annual period for the estimation of returns is consistent with the market valuation literature (Easton and Harris, 1991; Easton *et al.*, 1992; Feltham and Ohlson, 1995; Ohlson, 1995), extending over nine months prior to and three months after the fiscal year end. This reasoning is employed to include the influence of financial statement announcements, which public companies have to prepare and present by the end of the third month following the fiscal year end within returns.

*EARN*<sub>*it*</sub> denotes current earnings expressed as earnings per share for firm *i* in period *t* and is divided by the beginning of period stock price;  $\Delta EARN_{i,t}$  denotes the explanatory power that change in earnings may have over returns, expressed in the form of change in earnings per share for firm *i* in period *t* and further divided by the beginning of period stock price.

#### 3.2 Market perception of audit opinion types

This study's approach is based on a simplistic reasoning. It addresses audit opinion market effect by introducing to Equation (1) a series of dichotomous dummy variables,

142

thus generating the following equation:

$$R_{i,t} = a_0 + a_1 EARN_{i,t} + a_2 \Delta EARN_{i,t} + a_3 UNQOP_{i,t} + a_4 UNQEXP_{i,t} + a_5 QOP_{i,t} + e_{i,t}$$

This rationale attempts not only to capture the relationship between stock returns and audit opinion but also to thoroughly inspect the informative value of each distinct type of auditor report. In addition it aims to understand whether unqualified and qualified[3] auditor opinions can create a market effect. In other words, we model audit opinion expression in Equation (2) in order to examine the following hypothesis:

H1. Audit opinion types have incremental information content.

The categorisation of auditor opinion type used to create the dichotomous variables employed in Equation (2), is based on AU Section 508, Reports on Audited Financial Statements of SAS No. 58. The UNQOP variable represents Unqualified Opinion, depicting the type of opinion expressed by the auditor as verification that the financial statements fairly present a firm's financial position. This variable equals 1 if the audit report type falls in this category and 0 otherwise. The same fair representation of the firm's financial information applies to the Unqualified Opinion with Explanatory Paragraph, or UNQEXP[4]. The difference in this report type lies in the existence of certain circumstances that, while not affecting the audit opinion, forces the auditor to add an explanatory paragraph. UNQEXP is a dichotomous variable that equals 1 if the auditor report falls in this opinion type and 0 otherwise. QOP represents Qualified Opinion, demonstrating that the firm's financial information is fairly presented with the exception of the matter to which the qualification relates. This variable takes the value of 1 if the audit report type fits this description, and 0 otherwise (see Table AI for variable description). Statistically significant findings that may arise from estimating Equation (2) allow the identification of those report types that have a market effect whether qualified (QOP) or not (UNQOP, UNQEXP).

## 3.3 Market perception of audit opinion for big auditors and small, financially distressed auditees

To extend the analysis of market reactions to audit reports, this study introduces auditor size, auditee size and financial distress to Equations (1) and (2); these three factors constitute audit opinion determinants. These determinants were utilised because of their ability to create interactions with audit opinion types. This allows examination as to whether these intervening factors differentially impact the market perception of audit opinions.

Research suggests that there is a significant relationship between auditing firm size and audit opinion (Dye, 1993; Firth, 2002; Francis and Wang, 2008; Ireland, 2003; Krishnan and Stephens, 1996). This suggestion, combined with the fact that auditing firms' size has a positive relationship with audit quality (DeAngelo, 1981a, b; Francis and Yu, 2009; Ireland and Lennox, 2002), triggered our interest in empirically capturing the market impact of audit opinion types issued by a big-sized auditing firm. Large auditing firms seem to provide a higher quality of auditing services since they have more to lose by failing to report problems encountered, thus threatening their high reputation (DeAngelo, 1981a, b). In this context, it is only natural to explore whether opinion types reported by big auditing firms are validated by the investing public as value relevant. Accordingly, the following hypothesis is formulated:

*H2a*. Auditor size impacts the incremental information content of audit opinion.

Incremental information content

(2)

Following the same rationale, the utilisation of the relationship between the size of an audited firm and the audit opinion type (Citron and Taffler, 2004; Dopuch *et al.*, 1987; Firth, 2002; Ireland, 2003) enables us to conceptualise whether a specific audit opinion for a small auditee can have a substantial effect on investor decisions. It is implied that auditee size affects a firm's probability to receive a qualification (Afify 2009), therefore the smaller the firm, the more likely that a given uncertainty will be material (Francis and Krishnan, 1999; Monroe and Teh, 1993) and the lower the audit risk will be due to lower complexity and improved internal control (Jallow *et al.*, 2012). In other words, auditee size affects audit opinion, and small firms in particular tend to be negatively affected by the auditor's report issuance. However, a question arises as to whether this notion supports the market perception regarding the opinion types received by small auditees. This study conceptualises the interaction between auditee size and audit opinion types by investigating the following hypothesis:

H2b. Auditee size impacts the incremental information content of audit opinion.

The likelihood of a company's facing financial distress is also considered in the present analysis because the accounting literature regards it as a factor that significantly affects audit opinion (Ruiz-Barbadillo *et al.*, 2009; Carcello and Neal, 2000; Carey and Simnett 2006; Chi and Chin, 2011; Citron and Taffler, 1992; DeFond *et al.*, 2002; Hudaib and Cooke, 2005; Svanberg and Öhman, 2014). Taking this into account, the present study examines whether a firm's financial distress is perceived as important by the investing public when considering the informative value produced by various audit opinion types, leading to the following is hypothesis:

*H2c.* The likelihood of financial distress for a firm impacts the incremental information content of audit opinion.

With the introduction of the aforementioned factors, this study attempts a second level of analysis by extending Equation (2). These factors are incorporated as dichotomous variables and used to introduce their interactions with the various audit opinion types. Therefore, the addition of these new intercept variables allows one to observe how certain aspects of these factors impact market perception of audit opinion. The mathematical representation for this extended model is represented by following equation:

$$\begin{aligned} R_{i,t} &= a_0 + a_1 EARN_{i,t} + a_2 \Delta EARN_{i,t} + a_3 UNQOP_{i,t} + a_4 UNQEXP_{i,t} \\ &+ a_5 QOP_{i,t} + a_6 AUDITOR_{i,t} + a_7 SIZE_{i,t} + a_8 FD_{i,t} + a_9 UNQA_{i,t} \\ &+ a_{10} UNXPA_{i,t} + a_{11} QOPA_{i,t} + a_{12} UNQSZ_{i,t} + a_{13} UNXPSZ_{i,t} \\ &+ a_{14} QOPSZ_{i,t} + a_{15} UNQFD_{i,t} + a_{16} UNXPFD_{i,t} + a_{17} QOPFD_{i,t} + e_{i,t} \end{aligned}$$
(3)

This new equation incorporates *AUDITOR*, which represents audit firm size and is constructed as a dichotomous dummy variable that equals 1 if the firm performing the company's audit belonged to either Big 8, Big 6, Big 5 or Big 4 during 1981-2011, and 0 otherwise. *SIZE* is another dichotomous variable that represents the auditee's size. For its construction, the logarithm of the market value of equity is used as a size measure (Fama and French, 1995), and the audited firms are categorised into two distinct groups – small and big auditees – using the median size as an allocation measure. This variable has the value of 1 if the logarithm for each audited firm is smaller than the median logarithm (small firm) and 0 otherwise (big firm). *FD* denotes whether the audited firm has

IAAR

17.2

144

characteristics of financial distress. For *FD*, Altman's *Z*-score (Altman, 1968) is used as a measure of financial distress. The variable equals 1 if Altman's *Z*-Score for each audited firm is smaller than 1.81, thus representing a financially distressed firm, and 0 otherwise.

The intercept variables depicted in Equation (3) constitute the means by which to examine whether the intervening factors – *AUDITOR*, *SIZE* and *FD* – differentially impact the market perception of audit opinions. Their construction is straightforward because they comprise the interaction among the three intervening factors (*AUDITOR*, *SIZE* and *FD*) and audit opinion types (*UNQOP*, *UNQEXP* and *QOP*). Therefore, the intercept variables *UNQA*, *UNXPA* and *QOPA* are the products of *UNQOP*, *UNQEXP* and *QOP*, respectively, with *AUDITOR* as the dichotomous variable. Each of them represents the three audit opinion types issued solely by big auditing firms. In the same way, *UNQSZ*, *UNXPSZ* and *QOPSZ* comprise the products of the three audit opinion types issued for small auditees. Finally, the three audit opinion types, issued for financially distressed auditees, are depicted as *UNQFD*, *UNQEXP* and *QOPFD*, which constitute the respective products of the audit opinion types (*UNQOP*, *UNQEXP* and *QOP*) with *FD* as the dichotomous variable.

The potential statistical significance of the intercept variables included in the extended model enables better understanding of investor behaviour towards audit opinion, thus demonstrating the effects of certain audit opinion determinants on it. Furthermore, by comparing the two models (Equations (2) and (3)), useful information is obtained, concerning whether the informative value of our initial model is increased. Such a comparative analysis can provide useful insights not only for our analysis but also for the returns valuation discussion.

3.4 Market perception of audit opinion – introducing growth and leverage control variables To ensure the robustness and validity of the estimation results, this study includes in both models (initial and extended) certain proxies to act as controls in the relationship between investor decision reflected in stock returns and audit opinion. More specifically, proxies for growth opportunities and leverage are introduced in Equations (2) and (3). Growth has been a highly utilised control variable for returns or earnings relationships, representing the value of a firm's future investment opportunities and a significant relationship with stock returns (Collins and Kothari, 1989; Chen and Zhang, 2007). Leverage is a control variable that captures the effect of a long-term capital structure (Tsui *et al.*, 2001) and can be associated with returns either positively (Bhandari, 1988; Dhaliwal *et al.*, 2006) or negatively (Korteweg, 2010; Penman *et al.*, 2007). The inclusion of these control variables in both models allows the elimination of potential effects that these proxies might have on returns or audit opinion types and the returns-earnings relationship.

Furthermore, the growth and leverage proxies are used to capture the interaction between these control variables and the three distinct types of audit opinion. In other words, a series of slope variables is constructed to control for potential impact on audit opinion and introduced to Equation (3), in order to achieve a more robust estimation. The mathematical representation of the initial model, Equation (2), with the control variables is depicted in following equation:

$$R_{i,t} = a_0 + a_1 EARN_{i,t} + a_2 \Delta EARN_{i,t} + a_3 UNQOP_{i,t} + a_4 UNQEXP_{i,t} + a_5 QOP_{i,t} + a_6 GROWTH_{i,t} + a_7 LEV_{i,t} + e_{i,t}$$
(4)

The *GROWTH* proxy is constructed as the ratio of a firm's market value of equity to its book value of equity, implying that the higher the ratio, the greater the value of

growth opportunities. *LEV*, the proxy for leverage is calculated as the ratio of the firm's long-term debt to its total assets, thus capturing the portion of its assets financed by long-term debt.

Equation (5) allows for an examination of the extended model's robustness because it comprises the mathematical representation of Equation (3) with the addition of the two controls and their interactions with audit opinion types. Equation is as follows:

$$R_{i,t} = a_0 + a_1 EARN_{i,t} + a_2 \Delta EARN_{i,t} + a_3 UNQOP_{i,t} + a_4 UNQEXP_{i,t} + a_5 QOP_{i,t} + a_6 AUDITOR_{i,t} + a_7 SIZE_{i,t} + a_8 FD_{i,t} + a_9 UNQA_{i,t} + a_{10} UNXPA_{i,t} + a_{11} QOPA_{i,t} + a_{12} UNQSZ_{i,t} + a_{13} UNXPSZ_{i,t} + a_{14} QOPSZ_{i,t} + a_{15} UNQFD_{i,t} + a_{16} UNXPFD_{i,t} + a_{17} QOPFD_{i,t} + a_{18} GROWTH_{i,t} + a_{19} LEV_{i,t} + a_{20} UNQG_{i,t} + a_{21} UNXPG_{i,t} + a_{22} QOPG_{i,t} + a_{23} UNQL_{i,t} + a_{24} UNXPL_{i,t} + a_{25} QOPL_{i,t} + e_{i,t}$$
(5)

The UNQG, UNXPG and QOPG proxies denote the interaction between the GROWTH control variable and the audit opinion types (UNQOP, UNQEXP and QOP). Therefore, slope variables UNQG, UNXPG and QOPG are the respective products of UNQOP, UNQEXP and QOP with the GROWTH proxy. Moreover, UNQL, UNXPL and QOPL are the products of the three audit opinion types with the LEV control variable.

#### 3.5 Market perception of audit opinion – period analysis

The sample comprises data collected for a 30-year span including economic events from 1981 to 2011. During this period, several major events had a profound effect on the stock market as well as the auditing profession and practice. For the analysis to be complete, it is imperative to explore the research question in the context of these important economic events. Thus, the sample is divided into subperiods in an attempt to capture the investing public's changing perception of audit opinions over time. To do so, Equations (2) and (3) are estimated, allowing for a robustness check with (4) and (5) for each of the subsamples created.

These subperiods are constructed using various economic events – perceived as benchmarks in recent financial history – as intersections. The first subsample covers the years 1981-1987, examining the period before the stock market crash in 1987, also known as the Black Monday Crash. The next subsample covers 1988-2002: this precedes some of the biggest corporate scandals (e.g. Enron), the burst of the dot-com bubble and of course, the Arthur Andersen collapse, which led to the Sarbanes-Oxley Act (the SOX Act) in 2002 that deemed fraud detection a priority (Digabriele, 2009). In the context of the market perception of audit opinions, comparing these two subperiods can lead to understanding the reasons behind skepticism of auditing services.

The next subsample covers the subperiod 2003-2006, which examines the years post-SOX Act but before the financial crisis. The comparison of the pre-SOX and post-SOX era can facilitate a better understanding of how the global economy was led into the financial crisis. Finally, the comparison between 2007 and 2009, (the financial crisis period) and 2010-2011 (the post-crisis era) can reveal certain interesting facts about the financial crisis and the market perception of audit opinion.

IAAR

17.2

#### 4. The sample

The sample covers the period 1981-2011 and includes all public companies listed in the New York Stock Exchange, the American Stock Exchange and the NASDAQ stock market. The selected data excludes the financial and insurance sectors due to comparison difficulties with the industrial sectors. All the variables used in this analysis were available in the Compustat North America database. In addition to Compustat's CS active file, the CS research file (comprising firms that merged, delisted or ceased to exist) was also included in the sample, to eradicate potential survivorship bias that the exclusive use of long surviving firms would create. The variable specifications are presented in the Table AI.

The selection procedure resulted in a sample of 3,109 listed firms with 96,379 firmyear observations. However, 55,048 firm-year observations were excluded because no auditing information was available[5]. From the remaining observations, 18 firm-year observations, constituting adverse opinions and disclaimers were excluded because of their small number. In addition, 59 firm-year observations were omitted because they constituted audited firms that, according to Compustat, were left unaudited. The final sample consisted of 41,254 firm-year observations. A stage-by-stage explanation of the sample selection process is presented in Table I.

Table I shows that "Disclaimer" and "Adverse Opinion" are far from being regarded as important in the analysis, with one and 17 firm-year observations, respectively; therefore, they were excluded from the estimation. Furthermore, returns, annual earnings per share and change in earnings per share, divided by the beginning of period price, growth and leverage were winsorized for a fraction of  $\pm 1.5$  per cent to limit the occurrence of outliers.

Table II provides descriptive statistics on selected variables of the sample.

Despite using certain sample selection criteria to limit the occurrence of outliers, some extreme values (high and low) still appear. Moreover, examining the descriptive statistics, Table II suggests that the *R* and  $\Delta EARN$  variables present leptokurtic frequency distributions that are more concentrated at the lower end, while the *EARN* variable presents a negatively skewed leptokurtic frequency distribution with a concentration of frequencies at the higher end.

Table III provides data on the industry sector categorisation of the sample, thus depicting the largest firm concentration in the construction and manufacturing sectors, while the smallest number of companies, based on a SIC code categorisation, comprise firms listed in agriculture, forestry and fishing.

#### 5. Empirical results

To avoid multicollinearity problems, the continuous variables in the data set were tested using Pearson's correlations. The matrix created (Table IV) demonstrates a

Sample's contents	Firm-year observations
Initial sample	96,379
Missing audit opinion observations	55,048
	41,331
(–) Disclaimer 1	,
(–) Adverse opinion 17	Table I
(–) Unaudited 59	77 Sample selection
Final sample	41,254 process table

JAAK 17.2	Variable	Mean	Median	SD	Max	Min
17,2	R	0.238	0.088	0.829	4.091	-0.849
	EARN	0.010	0.046	0.189	0.490	-0.818
	$\Delta EARN$	0.032	0.008	0.221	1.122	-0.594
	UNQOP	0.653	1	0.476	1	0
	UNQEXP	0.337	0	0.473	1	0
148	QOP	0.008	0	0.088	1	0
	<i>ĂUDITOR</i>	0.785	1	0.410	1	0
	SIZE	0.282	0	0.450	1	0
	FD	0.348	0	0.476	1	0
	UNQA	0.513	1	0.500	1	0
	UNXPA	0.264	0	0.441	1	0
	QOPA	0.007	0	0.081	1	0
	ÜNQSZ	0.504	1	0.500	1	0
	UNXPSZ	0.229	0	0.420	1	0
	QOPSZ	0.006	0	0.079	1	0
	UNQFD	0.192	0	0.394	1	0
	UNXPFD	0.148	0	0.355	1	0
	QOPFD	0.005	0	0.072	1	0
	GROWTH	3.693	2.129	5.107	32.170	0.307
	LEV	25.810	20.820	27.230	148.300	0
	UNQG	2.518	1.339	4.503	32.170	0
	UNXPG	1.238	0	3.615	32.170	0
	QOPG	0.027	0	0.620	32.170	0
	UNQL	15.260	2.069	22.090	148.300	0
	UNXPL	10.730	0	24.830	148.300	0
	QOPL	0.299	0	4.501	148.300	0
	Noton The som	mla compilate of 41	954 firms many above	motions from the	nomial of 1001 2011	All the det

**Notes:** The sample consists of 41,254 firm-year observations from the period of 1981-2011. All the data were available by Compustat and refer to firms listed in NYSE, AMEX and NASDAQ. R, EARN,  $\Delta EARN$ , GROWTH and LEV have been winsorized for a fraction of  $\pm 1.5$  per cent. For variable specification see Table AI

Table II.	
Summary	statistics

	SIC code	SIC code description	Companies
	0	Agriculture, forestry and fishing	4
	1	Mining	120
	2	Construction	271
	3	Manufacturing	394
	4	Transportation, communications, electric, gas and sanitary services	203
	5	Wholesale and retail trade	96
	7	Hotel, personal, business, repair, A&R and motion picture services	170
	8	Health, legal, educational, social, museum, engineering and management	
Table III		services	66
Sample companies	99	Non-classifiable establishments	7
according to SIC		Total	1,333
code categorisation	Note: SI	C code represents the industry sector according to the SIC code categorisation	

positive relationship among scaled earnings, scaled earnings changes and growth with returns. These rather low and statistically significant correlation coefficients indicate a low possibility of the existence of multicollinearity. Returns also present a negative and statistically significant correlation with leverage that can again be characterised as low. For the empirical analysis, a fixed effects regression was implemented using industry- and year-specific effects for the estimation of Equations (2)-(5). This method allows for the fixed industry effects to identify the effect of otherwise unobservable industry characteristics that are stable over time and present a correlation with explanatory variables (Greene, 2000). Furthermore, the incorporation of fixed year effects was preferable to an alternative cross-sectional analysis for each year of the sample period which would have provided inconsistent results and little take away[6] because of the fragmentation of the sample into 31 smaller ones. This study's choice of estimation is consistent with the results of the F-test for the joint significance that was implemented, which resulted in favour of incorporating industry- and year-fixed effects (Kousenidis *et al.*, 2013).

#### 5.1 Market perception of audit opinion – full sample multivariate analysis

The initial regression analysis focuses on the estimation of the first two models: Equations (2) and (3). The outcomes of the estimated panel regressions with industryand year-fixed effects for these models are presented in Table V.

The results of Equation (2) show positive and statistically significant coefficients for the intercept and the independent variables *EARN* and *ΔEARN*. The significance level is 1 per cent for both coefficients and the intercept, providing consistent findings with the multivariate analysis of Easton and Harris' (1991) model. The findings from the three dichotomous variables representing the three types of audit opinion utilised (*UNQOP*, *UNQEXP* and *QOP*) are quite interesting. Specifically, *UNQOP*, *UNQEXP* and *QOP* are insignificant, presenting the possibility that the investing public, seems to be unaffected by the three types of audit opinions. This initial finding is in line with several market-based studies (Ameen *et al.*, 1994; Dodd *et al.*, 1984); however, a final conclusion cannot be drawn merely on the basis of the first estimation. The adjusted  $R^2$ estimation is 0.215, presenting a rather high degree of explanatory power for the initial model compared with the Easton and Harris results.

The findings from the industry- and year-fixed effects regression estimation of the extended model, Equation (3), are presented in Table V. Here, the incorporation of specific factors or determinants of audit opinions (*AUDITOR, SIZE* and *FD*) and their interactions with audit opinion types (*UNQOP, UNQEXP* and *QOP*), which constitute the intercept variables (*UNQA, UNXPA, QOPA, UNQSZ, UNXPSZ, QOPSZ, UNQFD, UNXPFD* and *QOPFD*), provide certain noteworthy estimation outcomes. Once more, the coefficients of *EARN* and  $\Delta EARN$  appear to be significant, presenting a positive relationship with stock returns. The *FD* dichotomous variable presents a statistically significant coefficient, while the newly introduced *AUDITOR* and *SIZE* dichotomous variables, representing big auditing firms and small audited firms, appear to be statistically insignificant for this estimation. The negative sign for the *FD* variable

	R	EARN	$\Delta EARN$	GROWTH	LEV	
R EARN ∆EARN GROWTH LEV Note: ***Signifi	1.000 $0.088^{***}$ $0.097^{***}$ $0.113^{***}$ $-0.056^{***}$ icant at the 0.01 le	1.000 0.311*** 0.134*** -0.171*** vel (two tailed)	1.000 0.048*** 0.031***	1.000 -0.021***	1.000	Table IV.Pearson's correlation matrix for continuous variables

JAAR 17.2		Equation (2)	Equation (3)	Equation (4)	Equation (5)
11,2	Intercept	0.388 (2.634)***	1.304 (2.496)**	0.536 (1.798)*	5.302 (2.132)**
	EARN	0.240 (5.056)***	0.158 (2.952)***	0.437 (6.800)***	0.289 (3.952)***
	$\Delta EARN$	0.167 (4.362)***	0.236(5.614)***	0.194 (3.963)***	0.260 (4.975)***
	UNQOP	$0.233 (1.779)^{*}$	-0.566(-1.099)	0.108 (0.331)	0.064 (0.281)
150	UNQEXP	0.192(1.463)	-0.593(-1.150)	0.086 (0.264)	0.066 (0.286)
150	QOP	0.123 (0.906)	-0.705(-1.286)	0.003(0.010)	-0.528(-0.022)
	AUDITOR	01120 (01000)	-0.701(-1.633)	010000 (01010)	4.768 (2.855)***
	SIZE		0.028 (1.017)		5.932 (3.040)***
	FD		-0.553 (-2.260)**		-0.885 (-9.110)***
	UNQA		0.645(1.502)		4 821 (-2 890)***
	UNXPA		0.650(1.512)		4.830 (-2.893)***
	QOPA		-0.769 (-1.683)*		-4.751 (-2.822)***
	UNQSZ		0.032 (2.863)***		5.915 (3.029)***
	UNXPSZ		0.029 (2.210)**		5.926 (3.036)***
	QOPSZ		-0.021 (-1.852)*		(omitted)
	UNQFD		0.381 (1.557)		0.723 (7.606)***
	UNXPFD		0.377 (1.538)		0.735 (7.692)***
	QOPFD		0.427 (1.645)*		0.805 (5.307)***
	GROWTH			0.020 (14.624)***	0.645 (3.263)***
	LEV			-0.002 (-6.062)***	-0.003 (-2.258)**
	UNQG				0.622 (3.145)***
	UNXPG				0.631 (3.189)***
	QOPG				0.645 (3.262)***
	UNQL				-0.003 (-2.597)***
	UNXPL				-0.002 (-1.849)*
	QOPL				-0.003(-1.057)
Table V.	Adj. $R^2$	0.215	0.224	0.242	0.248
Panel regression	Akaike criterion	73,694.961	68,652.963	60,700.645	57,672.447
estimation using	n	33,537	31,428	29,074	27,570
industry and year	Notes: Depender	t variable: Returns	(R); values in the br	ackets represent the t-	statistic value; coeffi-
fixed effects for the	cient values omitt	ed are dropped fror	n the estimation due	to collinearity; the sam	ple consists of 41,254
analysis of market	IIrm-year observa	tions from the peri	00 01 1981-2011. All	the data were availab	le by Compustat and
perception of audit	refer to firms liste	a in NYSE, AMEX a	and NASDAQ. R, EAL	KIV, ΔΕΑΚΙΝ, GROWTT	and LEV have been
opinion (full sample	winsorized for a	traction of $\pm 1.5$ p	er cent. For variabl	e specification see Ta	able AI. *** $p < 0.01;$

opinion (full sample 1981-2011)

\*\**p* < 0.05; \**p* < 0.1

coefficient seems to concur with the expectation that a financially distressed firm would imply lower market returns.

In this estimation, the coefficients for audit opinion type variables (UNQOP, UNQEXP and QOP) are statistically insignificant, similar to the results of the empirical analysis of Equation (2). However, the intercept variables capturing market's perception of audit opinion under a specific context present certain noteworthy findings. The unqualified types of audit opinion issued by big auditing firms (UNQA and UNXPA) are statistically insignificant, thus implying an absence of market effect. On the other hand, qualified opinion type issued by big size auditors (QOPA) bears statistical significance (10 per cent) and produces a negative coefficient sign. This finding suggests that qualified audit opinion issued by a big auditing firm is perceived as valid by the market, which is then negatively affected by it. Investors recognise the credibility of a qualification resulting from a high-quality auditing process (DeAngelo, 1981a, b), while unqualified opinions seem to be anticipated and market irrelevant.

The same applies to the intercept variables that illustrate the audit expression for financially distressed auditees. The unqualified types of opinion issued for these firms (UNQFD and UNXPFD) are statistically insignificant, while QOPFD, depicting qualified opinions, has a 10 per cent level of statistical significance. This shows that investors perceive audit opinions to be of informative value when issued for companies in a questionable financial state. The focal point in this case is the coefficient sign of the qualified opinion (QOPFD). While both the unqualified audit opinions in this case present no real market impact, the qualified opinion positively affects market perception. One can only speculate about an explanation for such a finding. Perhaps, despite the qualified audit report, the market perceives such a company to be in the worst situation; thus, a report of this type, opposed to a disclaimer or an adverse opinion, might as well be considered "good news".

The estimation results that are designed to capture the market impact of audit opinion types related to small auditees (UNQSZ, UNXPSZ and QOPSZ) present a statistically significant relationship with stock returns (1, 5 and 10 per cent, respectively). More specifically, unqualified opinion types present a positive market impact, while the estimation of qualified opinion presents a negative and statistically significant relationship with stock returns, thus implying a negative market perception of such an opinion type when expressed about small firms. These findings demonstrate that both qualified and unqualified audit reports for small firms affect investors by providing information that influences their estimations in terms of either the amount or riskiness of future cash flows (Choi and Jeter, 1992). The data implies that audit expression for small auditees is more likely to provide previously unavailable and unexpected information that can trigger a market reaction because the investing public follows big firms far more closely (Bhushan, 1989; Brennan and Hughes, 1991; Lang and Lundholm, 1996).

The adjusted  $R^2$  of the Equation (3) estimation equals 0.224, demonstrating an improvement from the addition of the various dichotomous and intercept variables. Table V provides a smaller Akaike criterion value of 68,652.963, presenting an improvement in the comparison of this extended value relevance model, Equation (3), with the initial model, Equation (2). These findings provide initial evidence to suggest that the incorporation of certain statistically significant dummy variables adds an incremental value to stock return, thus allowing us not to reject, at least partially, the initial Hypothesis *H1*. Furthermore, these results support the argument that certain types of auditor opinion as expressed under a specific context (*QOPA*, *UNQSZ*, *UNXPSZ*, *QOPSZ* and *QOPFD*) affect market perception, therefore allowing us to provide support for *H2a-H2c*.

#### 5.2 Audit opinion and the introduction of growth and leverage as control variables

To extend the estimation and ensure robust results, two additional Equations (4) and (5), were introduced to the methodological approach. The results from the industry regressions for Equations (4) and (5) estimated with fixed effects are also presented in Table V. Examining the empirical results of Equation (4) and comparing them with Equation (2), it is obvious that the introduction of the two proxies in this model, *GROWTH* and *LEV*, generates certain interesting changes. In terms of explanatory variables, the *EARN* and  $\Delta EARN$  variables remain statistically significant, thus validating once more the model's consistency with the findings of Easton and Harris (1991). The *GROWTH* and *LEV* control variables, introduced as proxies to ensure robustness, also bear some statistical significance, demonstrating a positive and

negative relationship, respectively, with returns. Consistent with the estimation of Equation (2), the UNQOP, UNQEXP and QOP variables are insignificant, demonstrating that the investing public, seems to be unaffected by the three types of audit opinions. Nevertheless, a noteworthy point can be traced in the differences between Equations (2) and (4).

Their comparison reveals a drop in the intercept's statistical significance from Equation (2) to Equation (4), a finding that can perhaps be explained by the implied non-zero effect, which the intercept has on all three equations. In addition, the increase in the adjusted  $R^2$  of Equation (4) is 0.242, from 0.215 in the case of Equation (2); this demonstrates an improvement from the addition of the control variables growth and leverage. Table V provides a smaller Akaike criterion figure, equal to 60,700.645, contrary to 73,694.961 in the case of Equation (2). Thus, another improvement is evident in the comparison of the evidently more robust model (Equation (4)) with the initial model (Equation (2)). Thus, it seems obvious that the introduction of specific proxies in the empirical analysis generate an improved model, allowing us to better understand the market perception of audit opinions by achieving more robust results. This supports the choice to incorporate these proxies and their interactions with audit opinion types in the extended model of Equation (3), thus creating Equation (5).

The estimation results of Equation (5) which are presented in Table V, illustrate certain interesting comparisons. The introduction of the two variables *GROWTH* and *LEV*, along with the proxies that show their interactions with audit opinion types (*UNQG*, *UNXPG*, *QOPG*, *UNQL*, *UNXPL* and *QOPL*), control for any potential effects that might otherwise affect the outcomes in the empirical analysis.

The findings from the industry- and year-fixed effects regression estimation of Equation (5) show statistically significant *EARN* and  $\Delta EARN$  coefficients; this finding is consistent with the estimation results of Equation (3). The *GROWTH* and *LEV* control variables appear to be statistically significant at the 1 and 5 per cent level, respectively. The *GROWTH* proxy illustrates a positive relationship with stock returns, while the *LEV* control variable demonstrates an inverse relationship with the dependent variable *R*. The slope variables *UNQG*, *UNXPG*, *QOPG*, *UNQL* and *UNXPL*, which act as proxies, appear to be statistically significant, while *QOPL* seems to have an insignificant relationship with market returns.

The *AUDITOR* dichotomous variable presents a positive and statistically significant relationship with the dependent variable, thus implying a direct (positive) relationship between auditing firm's size and returns. There seems to be a positive market reaction if a big auditor performs the audit, perhaps due to the relation between auditor size and audit quality (Walker, 2013). The results of this extended model estimation for the *SIZE* variable also seem consistent with the existence of the size anomaly (Fama and French, 1993, 1995). The positive and statistically significant relationship between small-sized firms and returns seem to verify Fama and French's finding that small-sized firms have room for growth, implying increased future returns. The *FD* dichotomous variable, consistent with the findings of Equation (3), is also statistically significant. Again, the negative sign implies lower market returns for financially distressed firms.

A finding that should be analysed even further is the statistical significance of certain Equation (5) variables reflecting audit opinions that are used to describe and investigate the relationship between the investing public and its perception of audit opinion informative value. Table V shows that all audit opinion types (UNQOP, UNQEXP and QOP) are once again statistically insignificant, showing no relationship between audit

opinion and stock returns. This does not imply that the market perceives these types of opinions as lacking value relevance. Perhaps this can be explained by the fact that market perception can be altered by certain audit opinion types, but only under a specific context. This is captured by incorporating the intercept variables in the model.

More specifically, as shown in Table V, unqualified audit opinion types (UNQA and UNXPA) and qualified types (QOPA) issued by big auditing firms are statistically significant and therefore, affect market returns. Unqualified opinions demonstrate a positive effect on investor's perception when issued by big-sized auditors. However, the findings present a negative relationship between returns and qualified audit opinions issued by big auditing firms, an outcome that validates the negative market impact such opinion types might have. This suggests that "big" auditing firm's audit expression, (qualified or not) will always function as news for investors because of their high quality, triggering a market effect that translates into a stock return reaction.

The intercept variables representing the interaction of auditee size with the various types of audit opinions produce similar results, at least for the unqualified opinion types, (*QOPSZ* is excluded from the estimation because of certain collinearity issues). The *UNQSZ* and *UNXPSZ* intercept variables are positively related to stock returns with a statistical significance level of 1 per cent. This finding shows that the market validates unqualified audit opinion issued for small firms by attributing positive value to them. It provides persistent evidence of the importance of size, and the manner in which an auditor's opinion impacts market's perception.

The UNQFD, UNXPFD and QOPFD intercept variables, which represent audit opinion types for financially distressed firms, depict statistically significant coefficients at the 1 per cent level, demonstrating a market effect. These estimation results can perhaps be explained by the "good news" notion discussed in relation to Equation (3). Unqualified opinions issued for firms facing serious financial challenges (UNQFD and UNXPFD) seem to positively impact the investing public; perhaps a positive opinion of an underperforming and financially distressed firm signals a positive change of pace to the market. Similarly, the peculiarity of QOPFD in Equation (3) persists in Equation (5), which also depicts a positive sign for this intercept variable's coefficient. The absence of a disclaimer or adverse opinion, or the issuance of a qualified opinion, can signify more encouraging prospects for the challenged firm, which might reflect its subsequent positive market impact.

The estimation of Equation (5) is a significant improvement over that of Equation (3). Comparing the two equations' results, it is evident that the addition of the proxies provides us with more precision because of a simultaneous increase in the adjusted  $R^2$  (from 0.224 to 0.248) and the decrease in the Akaike criterion figure (from 68,652.963 to 57,672.447). These results signify the increased ability of Equation (5) to generate answers regarding the market's perception of audit opinion. Estimating this extended model gives us the opportunity not only to check the robustness of the methodological rationale, but also to select a more complex empirical model to address audit opinion informative value. Furthermore, consistent with estimation without the proxies, the same results are achieved, allowing us to reject neither *H1*, nor *H2a*, *H2b* or *H2c*.

#### 5.3 Multivariate analysis of subperiods

Our sample, which covers a period of about 30 years (1981-2011), incorporates a large number of economic events that affected the capital market. Equations (2)-(5) were estimated for a series of subsamples representing subperiods to explore the research question in the context of these economic events. Table VI presents the results of these estimations for the first three subperiods: 1981-1987, 1988-2002 and 2003-2006. This

JAAR 17,2	Equation (5)	0.558***	(7.332) 0 136	(-0.725)	0.550***	(4.104) 0.027	(0.295)	(omitted)		(omitted) -	$-0.228^{***}$	(-3.841)	-0.050	(-1.367)	-0.267***	(-4.702) 0.030	(0.497)	(omitted)	I	(omitted)	- (omitted)	-	$-0.133^{***}$	(-2.915)	(omitted)	1.	(omitted)	0 111*	(1.654)	continued)
154	2006 Equation (4)	0.239***	(9.438) 0.013	(0.077)	0.479***	(0.179***	(7.473)	(omitted)		(omitted) -																				9
	2003- Equation (3)	-0.190	(-1.217) -0.415***	(-2.868)	0.413***	(nco.c) (0698***	(4.102)	0.640***	(666.4)	(Denuced) -	$-0.201^{***}$	(-3.953)	$0.091^{***}$	(3.031)	-0.217***	(-6.204) 0.036	(0.517)	(omitted)	I	(omitted)	_ 0117**	(2.545)	(omitted)	I	(omitted)	I	-0.086	()TCTL)	(Dammo)	
	Equation (2)	$-0.451^{***}$	(-3.132) -0.161	(-1.372)	0.285***	(0.888***	(6.113)	0.690***	(4.770)	(omittea)																				
	Equation (5)	-0.103	(-0.747) 0.420***	(3.746)	-0.053	(-0.07 /) 0.460***	(3.189)	0.409***	(6007)	(omitted)	0.105	(0.914)	$0.062^{***}$	(2.719)	-0.027	(-0.121)	(-1.153)	-0.184	(-1.387)	0.035	(otero) 0 157***	(5.585)	(omitted)	I	(omitted)	I	-0.118	0109	-0.448)	~
	002 Equation (4)	-0.309	(-1.246) 0.555***	(5.534)	-0.105	(-1.442) 0.482*	(1.946)	0.511**	(200.2)	-0.110 (-0.430)																				
	1988-2 Equation (3)	1.122*	(1.905) 0 168*	(1.904)	-0.021	(-0.757 - 0.757	(-1.285)	-0.853	(7 <del>14</del> 71—)	(omitted) –	-0.378	(-0.854)	(omitted)	I	-0.918**	(-1.960)	(0.768)	0.346	(0.776)	1.053**	(0 111***	(6.571)	0.032	(1.465)	$-1.510^{**}$	(-2.468)	0.724 /1 E48\	(01-0-10)	(1.520)	~
	Equation (2)	0.066	(0.353) 0.319***	(4.135)	-0.057	(-0.379) 0.125	(0.669)	0.111	(060.0)	-0.41/ <sup>7</sup> (-1.945)																				
	Equation (5)	0.314	(0.893) 1 656***	(10.062)	-0.229	(-1.309) -0.228	(-0.639)	(omitted)		(Denutred)	-0.327	(220-)	-0.014	(-0.786)	-0.046*	(-1.707) 0.986	0.846)	(omitted)	I	(omitted)	_ (omitted)	-	(omitted)	I	0.075	(086:0)	(omitted)	(omittad)	- -	
	-1987 Equation (4)	$-0.176^{***}$	(3.002) 1.71.6***	(10.810)	-0.118	(-0.002) 0.253***	(4.294)	(omitted)		-0.2227																				
Table VI.	1981 Equation (3)	0.137***	(6.455) 1 239***	(9.461)	-0.110	(-0.097) 0.193**	(2.350)	(omitted)	- 0	0.230	(omitted)	ļ	-0.093	(-1.386)	-0.112	(-0.437)	(-1.342)	(omitted)	I	-0.195	(700.0-)	(-1.313)	(omitted)	I	(omitted)	I	0.068	(ouzuu)	- -	
Panel regression estimation using industry and year fixed effects for the englying of magicat	Equation (2)	0.070	(0.370) 1 303***	(10.975)	-0.035	(-0.299)	(0.553)	(omitted)		0.100																				
perception of audit opinion (subperiod analysis 1981-2006)		Intercept	FARN		$\Delta EARN$	UNGOP		UNQEXP		40h	A UDI TOR		SIZE		FD	1 MOA	UANO	UNXPA		QOPA	ZSONIT		<b>UNXPSZ</b>		QOPSZ		UNQFD	LINEDED	OINALLD	

	Equation (2)	1981 Equation (3)	1-1987 Equation (4)	Equation (5)	Equation (2)	1988 Equation (3)	-2002 Equation (4)	Equation (5)	Equation (2)	2003- Equation (3)	2006 Equation (4)	Equation (5)
QOPED		060.0		0.021		0.924*		(omitted)		(omitted)		(omitted)
<b>GROWTH</b>		(670°N)	0.031***	0.038***		(0+0'T)	0.022***	- 0.067**		I	0.015***	- 0.012***
			(5.149)	(5.132)			(9.703)	(2.176)			(4.651)	(2.592)
LEV			-0.000	-0.002			-0.003***	0.017***			-0.003***	-0.001
UNDG			(-0.528)	(-1.322) (omitted)			(-7.200)	(30.335) 0.089***			(-4.435)	(-0.503) (omitted)
				-				(2.868)				-
UNXPG				(omitted)				0.085***				-0.001
				I				(2.740)				(-0.166)
QOPG				$-0.035^{***}$				0.084***				(omitted)
				(-2.912)				(3.133)				I
UNQL				0.003				$-0.020^{***}$				(omitted)
				(1.579)				(-25.245)				I
UNXPL				(omitted)				$-0.018^{***}$				-0.001
				Ι				(-15.566)				(-0.352)
QOPL				0.004				$-0.029^{***}$				(omitted)
				(1.353)				(-7.608)				I
Adj. $R^2$	0.121	0.105	0.168	0.157	0.006	0.021	0.035	0.041	0.015	0.033	0.035	0.048
Akaike criterior	1 5,155.141	4,590.631	4,404.204	3,938.358 9.72.0	30,859.905	28,381.444 19.700	25,680.981	24,365.893	17,510.566	16,434.155	14,167.305 5 444	13,660.745 5 222
и	3,232	2,988	CZ6,Z	2,138	13,793	12,789	12,058	11,340	0,445	0,008	5,444	5,223
Notes: Depende	ant variable: Ret	urns (R); values	s in the brackets	represent the t	-statistic value;	coefficient values	s omitted, are dro	pped from the e	estimation due to	collinearity; the	sample consists	of 41,254 firm-
year observatio.	ns from the per	riod of 1981-20	11. All the data	were available	by Compustat	and refer to firr	ms listed in NYS	E, AMEX and	NASDAQ. R, E	4RN, AEARN, C	ROWTH and L	EV have been
winsorized for z	fraction of $\pm 1$ .	5 per cent. For	variable specifi	cation see Tab	le AI. *** $p < 0.0$	$11; **p < 0.05; *_p$	b < 0.1					

Incremental information content

155

Table VI.

table depicts the market's perception of audit opinion types, demonstrating the differential impact that specific aspects of audit opinions (e.g. unqualified opinion received by small auditees) have on the informative value according to the investing public. These three subperiods signify a major change in the financial reporting and auditing process not only through the Black Monday crash of 1987 but also, more importantly, through the regulatory change introduced with the SOX Act. It becomes obvious from the results that there seems to be an increasingly higher informative value of audit opinion over time. There is a noticeable change between pre-SOX and post-SOX. During 2003-2006, regulatory environment improvements protecting the investing public had a profound effect on market's perception of the auditing process. The results in Table VI demonstrate exactly this phenomenon.

Within the pre-SOX subperiods (1981-1987 and 1988-2002) certain findings are of note. The results regarding audit opinion types are rather vague, but adequate for obtaining a better understanding of these two subperiods. Unqualified opinion types have statistical significance, implying a positive relationship with returns. They present a specific trend demonstrating that especially before SOX, ((1987-2002), as shown in Equation (5)), ungualified types severely impacted market perception towards audit opinions. Furthermore, UNQSZ presents a strong positive and statistically significant relationship with returns, at least during 1988-2002, implying that small audittees, receiving an unqualified opinion, can significantly impact investor perception. A qualified opinion, on the other hand, presents statistical significance in certain cases; however, these findings are not very strong since they are not consistent throughout the model's estimation. For the periods constituting the pre-SOX era, despite the little take away from these estimations, it is clear that unqualified opinions play an important role in the market and positively impact the investing public. Finally, it must be noted that the UNQEXP dichotomous variable and the intercept variables produced by it, were dropped from the estimation for 1981-1987 because no data are available prior to 1988.

The results show that the reputation of audit opinion went unscathed for the subperiod following the SOX Act's implementation in 2002. This subperiod before the financial crisis presents statistically significant findings for the unqualified opinion types, thus implying their validation by the market and the attribution of informative value in the auditing process. One particular dichotomous variable, *AUDITOR*, stands out, as it seems to be negatively related to returns. This implies that the smaller the audit firm, the larger the returns for the company it audits. In other words, the investing public during this period appears to show some disbelief of the notion that a big auditing firm may provide a higher quality of auditing services and subsequently, empowers the concept that trust is with the "little guy".

Table VII presents a similar concept regarding how the market perceived audit opinion during and after the financial crisis. Again, the *AUDITOR* variable depicts a negative coefficient sign, and this inverse relationship with returns becomes positive only following the financial crisis (2010-2011). However, there are some interesting findings yet again for unqualified opinions. There seems to be a positive and statistically significant relationship with returns, and this trend remains consistent during the post-SOX period for these opinion types (*UNQOP* and *UNQEXP*) for Equations (2)-(5). Even during the financial crisis and thereafter, the market seems to have perceived unqualified opinions as highly important relating them with their investing decisions, and thereby validating the disclosed financial information.

Overall, the empirical analysis of the sample's subperiods do not allow us to reject *H1* or the second set of hypotheses, (*H2a*, *H2b* and *H2c*), demonstrating the consistency

156

Equation (5)	$\begin{array}{c} 0.437^{****} \\ (3.460) \\ -0.007 \\ (-0.039) \\ 1.083^{****} \\ (9.216) \\ -0.019 \\ (000000 \\ 000010 \\ (000100 \\ 0.006 \\ 0.001 \\ 0.006 \\ (0.139) \\ 0.006 \\ (0.139) \\ 0.006 \\ (0.139) \\ 0.001 \\ (0.711) \\ (00000 \\ 0.091 \\ (0.711) \\ (00000 \\ 0.091 \\ (0.711) \\ (00000 \\ 0.001 \\ (0.711) \\ (00000 \\ 0.001 \\ (0.711) \\ (00000 \\ 0.001 \\ (0.711) \\ (00000 \\ 0.001 \\ (0.711) \\ (00000 \\ 0.001 \\ (00000 \\ 0.001 \\ (00000 \\ 0.001 \\ (00000 \\ 0.001 \\ (00000 \\ 0.001 \\ (00000 \\ 0.001 \\ (00000 \\ 0.001 \\ (00000 \\ 0.000 \\ (0000 \\ 0.000 \\ (0$	Incremental information content
)11 Equation (4)	$\begin{array}{c} 3.574^{****} \\ (43.656) \\ 0.129 \\ (0.818) \\ 1.035^{****} \\ (-41.679) \\ -3.062^{****} \\ (-41.679) \\ -3.062^{****} \\ (-42.151) \\ (-42.151) \end{array}$	157
2010-20 Equation (3)	$\begin{array}{c} 0.582^{****} \\ (11.587) \\ -0.162 \\ (-1.087) \\ 1.054^{****} \\ (10.095) \\ 0mitted) \\ - \\ 0.054 \\ (-0.392) \\ 0mitted) \\ - \\ 0.383^{****} \\ (-2.392) \\ (0mitted) \\ - \\ 0.336^{****} \\ (-3.599) \\ 0.336^{****} \\ (-1.119) \\ 0mitted) \\ - \\ 0.136 \\ (-1.119) \\ (0mitted) \\ - \\ 0.090 \\ (-1.002) \\ (0mitted) \\ (0mitted) \\ - \\ 0.000 \\ (-1.002) \\ (0mitted) \\ (0mitted) \\ - \\ 0.000 \\ (-1.002) \\ (0mitted) \\ - \\ 0.000 \\ (0mit$	
Equation (2)	$3.730^{****}$ (57.244) 0.013 0.097^{****} (10.013) $-3.127^{****}$ (-42.752) $-2.776^{*****}$ (-51.384)	
Equation (5)	0.047 (1.170) $0.592^{****}$ (3.380) $-0.358^{****}$ (-3.316) (omitted) - -0.3310 (omitted) - $-0.134^{****}$ (-5.054) (omitted) - $-0.090^{****}$ (-2.635) (omitted) - $-0.134^{****}$ (-2.635) (omitted) - $-0.134^{****}$ (-5.926) $-0.134^{****}$ (-5.926) $-0.134^{****}$ (-5.926) $-0.134^{****}$ (-5.926) $-0.134^{****}$	
2009 Equation (4)	$\begin{array}{c} -0.583^{****} \\ (-4.623) \\ 0.820^{****} \\ (5.591) \\ -0.452^{****} \\ (-4.664) \\ 0.3095) \\ 0.3095) \\ 0.507^{****} \\ (4.021) \\ (0mitted) \\ -\end{array}$	
2007- Equation (3)	$\begin{array}{c} 0.172^{****} \\ (5.018) \\ 0.211 \\ (1.598) \\ -0.205^{***} \\ (-2.389) \\ -0.062 \\ (-1.253) \\ (omitted) \\ - \\ -0.140^{****} \\ (-125.382) \\ (omitted) \\ - \\ -0.140^{****} \\ (-19.190) \\ 0.715^{****} \\ (47.586) \\ (omitted) \\ - \\ -0.138^{****} \\ (-7.878) \\ (omitted) \\ (omitted) \\ (omitted) \end{array}$	
Equation (2)	-0.662**** (-6.534) 0.451**** (4.409) -0.283**** (4.802) 0.489**** (4.802) 0.601**** (5.904) (omitted)	Table VII.           Panel regression           estimation using           industry and year
	Intercept EARN AEARN UNQOP UNQEXP QOP QOP SIZE ED UNQA UNXPA QOPA UNXPSZ UNXPSZ QOPSZ	analysis of market perception of audit opinion (subperiod analysis 2007-2011)

Downloaded by TAIPEI MEDICAL UNIVERSITY At 10:43 18 July 2016 (PT)

JAAR 17,2	Equation (5)	(omitted) – 0.007 (-0.063) (omitted) – 0.014**** (1.810) (0.005* (1.810) (0.005* (1.810) (0.0114* (1.810) (0.012* (1.810) (0.011* (-0.181) (0.011* (-0.181) (0.011* (-0.181) (0.011* (-0.181) (0.011* (-0.097 9,604.602 3,511 (1.748) 0.097 9,604.602 3,511 (1.748) 0.097 9,604.602 3,511 (1.748) 0.097 9,604.602 3,511 (1.748) 0.007 (0.011* (1.748) 0.007 (1.748) 0.007 (1.748) 0.007 (1.748) 0.007 (0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (0.011* (1.748) 0.007 (1.748) (1.748) 0.007 (1.748) (1.748) (1.748) (1.748) (1.748) (1.748) (1.748) (1.748) (1.748) (1.778) (1.748) (1
158	2011 Equation (4)	0.013**** (3.469) 0.002** (2.266) (2.266) (0.092 10,072.081 3,672 0.092 10,072.081 3,672 compustat and refe
	2010 Equation (3)	(omitted $)-0.012(-0.130)($ omitted $)-($ omitted $)-11,270.5914,032alues omitted are dwere available by (ction of \pm 1.5 per c$
	Equation (2)	0.078 11,823.322 4,215 alue; coefficient vi 2011. All the data insorized for a fra
	Equation (5)	$\begin{array}{c} \  \  \  \  \  \  \  \  \  \  \  \  \ $
	7-2009 Equation (4)	0.013**** (5.342) -0.001*** (-2.500) (-2.500) (-2.500) (-2.500) (-2.500) (-2.500) (-2.500) (-2.500) (-2.500)
	200' Equation (3)	$\begin{array}{c} & & & \\ & & & 0.025 \\ & & & 0.001 \\ & & & 0.001 \\ & & & 0.001 \\ & & & (0.024) \\ & & & (0.024) \\ & & & & (0.054) \\ & & & & & \\ & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & &$
	Equation (2)	0.030 9,918.014 5,852 14 variable: Return ample consists of and NASDAQ. <i>A</i> and NASDAQ. <i>A</i>
Table VII.		UNQFD UNXPFD QOPFD GROWTH LEV UNQG UNQG UNXPG QOPG UNXPL QOPL Adj. R <sup>2</sup> Adj. Adj. Adj. Adj. Adj. Adj. Adj. Adj.

Downloaded by TAIPEI MEDICAL UNIVERSITY At 10:43 18 July 2016 (PT)

of this study's findings and highlighting the notion that audit opinions bear incremental information content, even over a period spanning 30 years.

#### 6. Summary and conclusions

The deteriorating usefulness of financial information to investors (Lev and Zarowin, 1999), caused by the decline of financial reporting quality, has raised, questions about the validity of audit procedure on many occasions (Walker, 2013), while the evolution of financial reporting can challenge it even further in the future (Ismail and Sobhy, 2009). This paper provides empirical support for the incremental information content of audit opinion, thus supporting the reliability of the auditing process, which seems in most cases, to be irreplaceable in the minds of investors.

Our approach attempts to provide an in-depth analysis of a series of models by implementing a methodological rationale that has not been previously employed in the context of the audit opinion informational content. The analysis emphasises the examination of each audit opinion type in order to investigate which is perceived by the market as value relevant. Furthermore, this study verifies that a series of factors differentially impact the market perception of audit opinions and outlines the factor's interactions with audit opinion types. Moreover, to ensure robust findings, certain proxies are introduced. Their interactions with audit opinion types, aim to eradicate any potential effect on stock returns or audit opinion types. Finally, the constructed estimation models are devised for a series of subperiods by using important economic events as intersections.

The primary analysis shows that the market will attribute informative value to both unqualified and qualified opinion types connected with specific factors impacting audit opinions, such as auditor and auditee size or financial distress. Specifically, the market perceives as value relevant qualified audit opinions issued by big auditors to small and distressed auditeed firms, while unqualified opinions seem to present a market effect only when an audit opinion type is issued for a small auditee.

The interaction of control variables (growth and leverage) with audit opinion types as proxies provide equally interesting results verifying that audit opinions, both unqualified and qualified, generate a profound market effect. Unqualified opinion types by big auditing firms for small-sized and financially distressed auditees positively impact the market's perception. Similarly, as previously noted, a qualified opinion issued by big auditors for small auditees negatively affects the investing public, due to the seriousness of such an audit announcement. However, this does not seem to be the case with qualifications issued for financially distressed auditees. The positive market effect of this audit opinion type can perhaps be explained by the assumption that despite the firm's poor financial condition, it receives a qualified opinion. Thus it contradicts market's anticipation for a disclaimer or an adverse opinion, and acknowledges this qualification as an improvement.

The third part of the methodology analyses the market perception of auditor opinion over time. The findings demonstrate rather consistently that its importance remains unscathed by the passage of time. Despite the skepticism developed towards the auditing process (Paino *et al.*, 2012) and the subsequent criticism of the regulatory framework, the market seems to perceive unqualified opinions as valid, thereby attributing informational content to them. Such investors' behaviour seems to apply not only to the pre-SOX period but also post-SOX including, the financial crisis of 2007-2009 and thereafter.

This study, to the extent of our knowledge, is believed to be one of the first empirical analyses of audit opinion to employ a value relevance model. This contrasts with other market-based research adopting an event study methodology. It uses a 30-year US sample

to assess whether audit opinion types have a significant market effect and it examines the conditions under which each opinion has the most serious effect on investors. The research illustrates that both shareholders and potential investors use the information content of audit opinion for their investment decisions. The findings therefore suggest that investors rely on the validity of the audited, disclosed financial information.

This study also offers useful insights for auditors since it demonstrates the importance of the auditing profession to the users of financial statements. The information communicated through the auditing process represents comprehensive evidence, validating corporate disclosures (Tauringana and Chong, 2004), and ensuring the orderly operation of every financial environment. Furthermore, the value relevance of audit opinion and its resulting implied significance underlines the role of regulatory bodies and standard setters. The informative value of audit expression implies that regulators must establish with certainty that the framework both ensures and protects the proper implementation of the auditing process and allows them to track the efficiency of the various audit policy changes.

However, some research limitations exist in the analysis performed. Despite a rather large data set, an analysis regarding the disclaimer and adverse opinion was not possible. The small number of firm-year observations for these audit qualifications forced their omission. Nevertheless, this does not change the fact that the introduction of these variables and their interactions could add valuable attributes to this empirical research. The incorporation of these audit opinion types remains a topic for future research because the accumulation of additional years could allow for a higher number of observations, thus making their utilisation possible.

#### Notes

- 1. AU Section 339 "Audit Documentation", Section 23, para.482.
- "The issuance date can change if events occur after the completion of fieldwork, but before the issuance of the report" (AU Section 530 "Dating of the Independent Auditor's Report", Section 01, p. 641).
- 3. In this context, "qualified" is used to describe only a qualified opinion. The reason for not considering the introduction of a disclaimer and adverse opinion in the model is that this sample consists of merely 19 firm-year observations for both of these opinion types, rendering them practically unusable.
- 4. Unqualified opinion with an explanatory paragraph type is available from Compustat after 1988.
- 5. These firm-year observations cannot be classified as unaudited because Compustat does not provide us with any type of auditing information about them, including whether they are unaudited.
- 6. The anonymous reviewer is acknowledged here for suggesting this methodological rationale.

#### References

- Adelopo, I. (2009), "Modelling issues in the relationship between audit and non-audit fees", Journal of Applied Accounting Research, Vol. 10 No. 2, pp. 96-108.
- Afify, H. (2009), "Determinants of audit report lag: does implementing corporate governance have any impact? Empirical evidence from Egypt", *Journal of Applied Accounting Research*, Vol. 10 No. 1, pp. 56-86.

160

IAAR

- Al-Thuneibat, A.A., Khamees, B.A. and Al-Fayoumi, N.A. (2008), "The effect of qualified auditors' opinions on share prices: evidence from Jordan", *Managerial Auditing Journal*, Vol. 23 No. 1, pp. 84-101.
- Altman, E.I. (1968), "Financial ratios, discriminant analysis and the prediction of corporate bankruptcy", *The Journal of Finance*, Vol. 23 No. 4, pp. 589-609.
- Ameen, C., Elsie, C.M. and Guffey, M.D. (1994), "Information content of the qualified audit opinions for over-the-counter firms", *Journal of Business Finance & Accounting*, Vol. 21 No. 7, pp. 997-1011.
- Aziz, U.-F. and Omoteso, K. (2014), "Reinforcing users' confidence in statutory audit during a post-crisis period: an empirical study", *Journal of Applied Accounting Research*, Vol. 15 No. 3, pp. 308-322.
- Bailey, K.E., Bylinski, J.H. and Shields, M.D. (1983), "Effects of audit report wording changes on the perceived message", *Journal of Accounting Research*, Vol. 21 No. 2, pp. 355-370.
- Bailey, W.T. (1981), "The effects of audit reports on chartered financial analysts' perceptions of the sources of financial-statement and audit-report messages", Accounting Review, Vol. 56 No. 4, pp. 882-896.
- Ball, R.J., Walker, R. and Whittred, G. (1979), "Audit qualifications and share prices", ABACUS, Vol. 15 No. 2, pp. 23-34.
- Banks, D.W. and Kinney, W.R. Jr (1982), "Loss contingency reports and stock prices: an empirical study", *Journal of Accounting Research*, Vol. 20 No. 1, pp. 240-254.
- Barth, M.E., Beaver, W.H. and Landsman, W.R. (2001), "The relevance of the value relevance literature for financial accounting standard setting: another view", *Journal of Accounting* and Economics, Vol. 31 Nos 1/3, pp. 77-104.
- Baskin, E.F. (1972), "The communicative effectiveness of consistency exceptions", Accounting Review, Vol. 47 No. 1, pp. 38-51.
- Bessell, M., Anandarajan, A. and Umar, A. (2003), "Information content, audit reports and goingconcern: an Australian study", *Accounting and Finance*, Vol. 43 No. 3, pp. 261-282.
- Bhandari, L. (1988), "Debt/equity ratio and expected common stock returns: empirical evidence", *Journal of Finance*, Vol. 43 No. 2, pp. 507-528.
- Bhushan, R. (1989), "Firm characteristics and analyst following", Journal of Accounting and Economics, Vol. 11 Nos 2/3, pp. 255-274.
- Biddle, G.C., Hilary, G. and Verdi, R.S. (2009), "How does financial reporting quality relate to investment efficiency?", *Journal of Accounting and Economics*, Vol. 48 Nos 2/3, pp. 112-131.
- Brennan, M.J. and Hughes, P.J. (1991), "Stock prices and the supply of information", *The Journal of Finance*, Vol. 46 No. 5, pp. 1665-1691.
- Brito, R.D. and Peres, E.P. (2006), "Does credible auditing add value", *Brazilian Business Review*, Vol. 3 No. 2, pp. 200-222.
- Carcello, J.V. and Neal, T.L. (2000), "Audit committee composition and auditor reporting", *The Accounting Review*, Vol. 75 No. 4, pp. 453-467.
- Carey, P. and Simnett, R. (2006), "Audit partner tenure and audit quality", *The Accounting Review*, Vol. 81 No. 3, pp. 653-676.
- Chen, J.P. and Zhang, G. (2007), "How do accounting variables explain stock price movements? Theory and evidence", *Journal of Accounting and Economics*, Vol. 43 Nos 2/3, pp. 219-244.
- Chen, J.P., Charles, S.X. and Zhao, R. (2000), "An emerging market's reaction to initial modified audit opinions: evidence for the Shanghai Stock Exchange", *Contemporary Accounting Research*, Vol. 17 No. 33, pp. 429-455.

Chi, HY. and Chin	, CL. (2011),	, "Firm versus	partner r	measures o	of auditor	r industry	expertise	9
and effects or	n auditor qua	ality", Auditing	: A Journ	al of Pract	ice & Th	heory, Vol.	30 No. 2	,
pp. 201-229.								

- Choi, S.K. and Jeter, D.C. (1992), "The effects of qualified audit opinions on earnings response coefficients", *Journal of Accounting and Economics*, Vol. 15 Nos 2/3, pp. 229-247.
- Chow, C.W. and Rice, S.J. (1982), "Qualified audit opinions and share prices an investigation", Auditing: A Journal of Practice & Theory, Vol. 1 No. 2, pp. 35-53.
- Citron, D.B. and Taffler, R.J. (1992), "The audit report under going concern uncertainties: an empirical analysis", Accounting and Business Research, Vol. 22 No. 88, pp. 337-345.
- Citron, D.B. and Taffler, R.J. (2004), "The comparative impact of an audit report standard and an audit going-concern standard on going-concern disclosure rates", *Auditing: A Journal of Practice & Theory*, Vol. 23 No. 2, pp. 119-130.
- Collins, D. and Kothari, S.K. (1989), "An analysis of intertemporal and cross-sectional determinants of earnings response coefficients", *Journal of Accounting and Economics*, Vol. 11 Nos 2/3, pp. 143-181.
- Craswell, A.T. (1985), "Studies of information content of qualified audit reports", Journal of Business, Finance and Accounting, Vol. 12 No. 1, pp. 93-116.
- Czernkowski, R., Green, W. and Wang, Y. (2010), "The value of audit qualifications in China", Managerial Auditing Journal, Vol. 25 No. 5, pp. 404-426.
- Davis, R. (1982), "An empirical evaluation of auditors' 'subject-to' opinions", Auditing: A Journal of Practice and Theory, Vol. 2 No. 1, pp. 13-32.
- DeAngelo, L.E. (1981a), "Auditor size and audit quality", *Journal of Accounting and Economics*, Vol. 3 No. 3, pp. 183-199.
- DeAngelo, L.E. (1981b), The Auditor-Client Contractual Relationship: An Economic Analysis, UMI Research Press, Ann Arbor, MI.
- DeFond, M.L., Raghunandan, K. and Subramanyam, K. (2002), "Do non-audit service fees impair auditor independence? Evidence from going concern audit opinions", *Journal of Accounting Research*, Vol. 40 No. 4, pp. 1247-1274.
- Dhaliwal, D., Heitzman, S. and Zhen, L. (2006), "Taxes, leverage, and the cost of equity capital", *Journal of Accounting Research*, Vol. 44 No. 4, pp. 691-723.
- DiGabriele, J.A. (2009), "Implications of regulatory prescriptions and audit standards on the evolution of forensic accounting in the audit process", *Journal of Applied Accounting Research*, Vol. 10 No. 2, pp. 109-121.
- Dillard, J.F. and Jensen, D.L. (1983), "The auditor's report: an analysis of opinion", *The Accounting Review*, Vol. 58 No. 4, pp. 787-798.
- Dodd, P., Dopuch, N., Holthausen, R. and Leftwich, R. (1984), "Qualified audit opinions and stockprices: information content, announcement dates, and concurrent disclosures", *Journal of Accounting and Economics*, Vol. 6 No. 1, pp. 3-38.
- Dopuch, N., Holthausen, R.W. and Leftwich, R.W. (1986), "Abnormal stock returns associated with media disclosures of 'subject to' qualified audit opinions", *Journal of Accounting and Economics*, Vol. 8 No. 1, pp. 93-117.
- Dopuch, N., Holthausen, R.W. and Leftwich, R.W. (1987), "Predicting audit qualifications with financial and market variables", Accounting Review, Vol. 62 No. 3, pp. 431-454.
- Dye, R.A. (1993), "Auditing standards, legal liability and auditor wealth", Journal of Political Economy, Vol. 101 No. 5, pp. 887-914.
- Easton, P.D. and Harris, T.S. (1991), "Earnings as an explanatory variable for returns", *Journal of Accounting Research*, Vol. 29 No. 1, pp. 19-36.

JAAR 17.2

- Easton, P.D., Harris, T.S. and Ohlson, J.A. (1992), "Aggregate accounting earnings can explain most of security returns: the case of long return intervals", *Journal of Accounting and Economics*, Vol. 15 Nos 2/3, pp. 119-142.
- Elliott, J.A. (1982), "Subject to' audit opinions and abnormal security returns outcomes and ambiguities", *Journal of Accounting Research*, Vol. 20 No. 2, pp. 617-638.
- Estes, R. and Reimer, M. (1977), "A study of the effects of qualified auditors' opinions on bankers lending decisions", Accounting & Business Research, Vol. 7 No. 28, pp. 250-259.
- Fama, E.F. and French, K.R. (1993), "Common risk factors in the returns on stocks and bonds", Journal of Financial Economics, Vol. 33 No. 1, pp. 3-56.
- Fama, E.F. and French, K.R. (1995), "Size and book-to-market factors in earnings and returns", *The Journal of Finance*, Vol. 50 No. 1, pp. 131-155.
- Fargher, N.L. and Wilkins, M.S. (1998), "Evidence on risk changes around audit qualification and qualification withdrawal announcements", *Journal of Business Finance & Accounting*, Vol. 25 Nos 7/8, pp. 829-847.
- FASB (2010), "Conceptual Framework Statement of Financial Accounting Concepts No. 8", Financial Accounting Standards Board (FASB) of The Financial Accounting Foundation, Norwalk, CT.
- Feltham, G. and Ohlson, J.A. (1995), "Valuation and clean surplus accounting for operating and financing activities", *Contemporary Accounting Research*, Vol. 11 No. 2, pp. 689-731.
- Firth, M. (1978), "Qualified audit reports: their impact on investment decisions", Accounting Review, Vol. 53 No. 3, pp. 642-649.
- Firth, M. (1980), "A note on the impact of audit qualifications on lending and credit decisions", Journal of Banking & Finance, Vol. 4 No. 3, pp. 257-267.
- Firth, M. (2002), "Auditor-provided consultancy services and their associations with audit fees and audit opinions", *Journal of Business Finance & Accounting*, Vol. 29 Nos 5/6, pp. 661-693.
- Francis, J. and Wang, D. (2008), "The joint effect of investor protection and Big 4 audits on earnings quality around the world", *Contemporary Accounting Research*, Vol. 25 No. 1, pp. 157-191.
- Francis, J. and Yu, M. (2009), "Big 4 office size and audit quality", *The Accounting Review*, Vol. 84 No. 5, pp. 1521-1552.
- Francis, J.R. and Krishnan, J. (1999), "Accounting accruals and auditor reporting conservatism", *Contemporary Accounting Research*, Vol. 16 No. 1, pp. 135-165.
- Greene, W. (2000), Econometric Analysis, 4th ed., Prentice-Hall, Upper Saddle River, NJ.
- Guillamon, G. (2003), "The usefulness of the audit report in investment and financing decisions", Managerial Auditing Journal, Vol. 18 Nos 6/7, pp. 549-559.
- Gul, F.A. (1987), "The effects of uncertainty reporting on lending officers' perceptions of risk and additional information required", ABACUS, Vol. 23 No. 2, pp. 172-181.
- Harris, T.S. and Ohlson, J.A. (1987), "Accounting disclosures and the market valuation of oil and gas properties", *The Accounting Review*, Vol. 62 No. 4, pp. 651-670.
- Hatherly, D., Innes, J. and Brown, T. (1991), "The expanded audit report an empirical investigation", Accounting & Business Research, Vol. 21 No. 84, pp. 311-319.
- Herbohn, K., Ragunathan, V. and Garsden, R. (2007), "The horse has bolted: revisiting the market reaction to going concern modifications of audit reports", *Accounting & Finance*, Vol. 47 No. 3, pp. 473-493.
- Holt, G. and Moizer, P. (1990), "The meaning of audit report", Accounting & Business Research, Vol. 20 No. 78, pp. 111-121.

163

Incremental

information

JAAR 17.2	Hope, OK., Thomas, W.B. and Vyas, D. (2013), "Financial reporting quality of US private and public firms", Accounting Review, Vol. 88 No. 5, pp. 1715-1742.
	Houghton, K.A. (1983), "Audit reports: their impact on the loan decision process and outcome: an experiment", Accounting & Business Research, Vol. 14 No. 53, pp. 15-20.
164	Hudaib, M. and Cooke, T.E. (2005), "The impact of managing director changes and financial distress on audit qualification and auditor switching", <i>Journal of Business Finance &amp;</i> <i>Accounting</i> , Vol. 32 Nos 9/10, pp. 1703-1739.
	Innes, J., Brown, T. and Hatherly, D. (1997), "The expanded audit report – a research study within the development of SAS 600", <i>Accounting, Auditing &amp; Accountability Journal</i> , Vol. 10 No. 5, pp. 702-717.
	Ireland, J.C. (2003), "An empirical investigation of determinants of audit reports in the UK", Journal of Business Finance & Accounting, Vol. 30 Nos 7/8, pp. 975-1016.
	Ireland, J.C. and Lennox, C. (2002), "The large audit firm fee premium: a case of selectivity bias?", Journal of Accounting, Auditing & Finance, Vol. 17 No. 1, pp. 73-91.
	Ismail, T.H. and Sobhy, N.M. (2009), "Determinants of auditors' perceptions of the work needed in the audit of internet-based financial reports in Egypt", <i>Journal of Applied Accounting</i> <i>Research</i> , Vol. 10 No. 2, pp. 132-150.
	Ittonen, K. (2012), "Market reactions to qualified audit reports: research approaches", Accounting

- Research Journal, Vol. 25 No. 1, pp. 8-24. Jallow, K., Adelopo, I., Jallow, K. and Scott, P. (2012), "Multiple large ownership structure, audit
- committee activity and audit fees: evidence from the UK", Journal of Applied Accounting Research, Vol. 13 No. 2, pp. 100-121.
- Korteweg, A. (2010), "The net benefits to leverage", The Journal of Finance, Vol. 65 No. 6, pp. 2137-2170.
- Kothari, S.P. and Zimmerman, J.L. (1995), "Price and return models", Journal of Accounting and Economics, Vol. 20 No. 2, pp. 155-192.
- Kousenidis, D.V., Ladas, A.C. and Negakis, C.I. (2013), "The effects of the European debt crisis on earnings quality", International Review of Financial Analysis, Vol. 30, pp. 351-362.
- Krishnan, J. and Stephens, R.G. (1996), "The simultaneous relation between auditor switching and audit opinion: an empirical analysis", Accounting & Business Research, Vol. 26 No. 3, pp. 224-236.
- Landsman, W.R. and Magliolo, J. (1988), "Cross-sectional capital market research and model specification", Accounting Review, Vol. 63 No. 4, pp. 586-604.
- Lang, M.H. and Lundholm, R.J. (1996), "Corporate disclosure policy and analyst behavior", Accounting Review, Vol. 71 No. 4, pp. 467-492.
- Lev, B. and Zarowin, P. (1999), "The boundaries of financial reporting and how to extend them", Journal of Accounting Research, Vol. 37 No. 2, pp. 353-385.
- Libby, R. (1979), "Bankers' and auditors' perceptions of the message communicated by the audit report", Journal of Accounting Research, Vol. 17 No. 1, pp. 99-122.
- Lin, Z.J., Tang, Q. and Xiao, J. (2003), "An experimental study of users responses to qualified audit reports in China", Journal of International Accounting, Auditing and Taxation, Vol. 12 No. 1, pp. 1-22.
- Martinez, M.C.P., Martinez, A.V. and Benau, M.A. (2004), "Reaction of the Spanish capital market to qualified audit reports", European Accounting Review, Vol. 4 No. 2, pp. 261-280.
- Miller, J., Reed, S.A. and Strawser, R.H. (1993), "Bank loan officers' perceptions of the new audit report", Accounting Horizons, Vol. 7 No. 1, pp. 39-52.

- Mittelstaedt, H.F., Regier, P.R., Chewning, E.G. and Pany, K. (1992), "Do consistency modifications provide information to equity markets?", *Auditing: A Journal of Practice & Theory*, Vol. 11 No. 1, pp. 83-98.
- Monroe, G.S. and Teh, S.T. (1993), "Predicting uncertainty audit qualifications in Australia using publicly available information", Accounting & Finance, Vol. 33 No. 2, pp. 79-106.
- Numata, S. and Takeda, F. (2010), "Stock market reactions to audit failure in Japan: the case of Kanebo and ChuoAoyama", *The International Journal of Accounting*, Vol. 45 No. 2, pp. 175-199.
- Ohlson, J. (1995), "Earnings, book values and dividends in security valuation", *Contemporary* Accounting Research, Vol. 11 No. 2, pp. 661-687.
- Paino, H., Smith, M. and Ismail, Z. (2012), "Auditor acceptance of dysfunctional behaviour: an explanatory model using individual factors", *Journal of Applied Accounting Research*, Vol. 13 No. 1, pp. 37-55.
- Penman, S., Richardson, S. and Tuna, I. (2007), "Book-to-price effect in stock returns: accounting for leverage", *Journal of Accounting Research*, Vol. 45 No. 2, pp. 427-467.
- Power, M.K. (2003), "Auditing and the production of legitimacy", Accounting, Organizations and Society, Vol. 28 No. 4, pp. 379-394.
- Pucheta Martínez, M.C., Martínez, A.V. and García Benau, M.A. (2004), "Reactions of the Spanish capital market to qualified audit reports", *European Accounting Review*, Vol. 13 No. 4, pp. 689-711.
- Rajgopal, S. and Venkatachalam, M. (2011), "Financial reporting quality and idiosyncratic return volatility", *Journal of Accounting and Economics*, Vol. 51 Nos 1/2, pp. 1-20.
- Robertson, J.C. (1988), "Analysts' reactions to auditors' messages in qualified reports", Accounting Horizons, Vol. 2 No. 2, pp. 82-89.
- Ruiz-Barbadillo, E., Gomez-Aguilar, N. and Carrera, N. (2009), "Does mandatory audit firm rotation enhance auditor independence? Evidence from Spain", *Auditing: A Journal of Practice & Theory*, Vol. 28 No. 1, pp. 113-135.
- Slack, R., Shrives, P., Rowbottom, N. and Lymer, A. (2010), "Exploring the use and users of narrative reporting in the online annual report", *Journal of Applied Accounting Research*, Vol. 11 No. 2, pp. 90-108.
- Soltani, B. (2000), "Some empirical evidence to support the relationship between audit reports and stock prices – the French case", *International Journal of Auditing*, Vol. 4 No. 3, pp. 269-291.
- Svanberg, J. and Öhman, P. (2014), "Lost revenues associated with going concern modified opinions in the Swedish audit market", *Journal of Applied Accounting Research*, Vol. 15 No. 2, pp. 197-214.
- Tauringana, V. and Chong, G. (2004), "Neutrality of narrative discussion in annual reports of UK listed companies", *Journal of Applied Accounting Research*, Vol. 7 No. 1, pp. 74-107.
- Tsui, J., Jaggi, B. and Gul, F. (2001), "CEO domination, growth opportunities, and their impact on audit fees", *Journal of Accounting, Auditing & Finance*, Vol. 16 No. 3, pp. 189-208.
- Walker, M. (2013), "How far can we trust earnings numbers? What research tells us about earnings management", Accounting and Business Research, Vol. 43 No. 4, pp. 445-481.

#### Further reading

AICPA (1988), "Reports on audited financial statements", Statement on Auditing Standards (SAS) No. 58, AU section 508, American Institute of Certified Public Accountants (AICPA), New York, NY.

JAAR 172	AICPA (2005), "Audit Documentation Statement on Auditing Standards (SAS) No. 103", American Institute of Certified Public Accountants (AICPA), New York, NY.
11,2	Amir, E., Harris, T.S. and Venuti, E.K. (1993), "A comparison of the value-relevance of US versus non-US GAAP accounting measures using form 20-F reconciliations", <i>Journal of</i> <i>Accounting Research</i> , Vol. 31, Supplement, pp. 230-264.
166	Beaver, W., Clarke, R. and Wright, W.F. (1979), "The association between unsystematic security returns and the magnitude of earnings forecast errors", <i>Journal of Accounting Research</i> , Vol. 17 No. 2, pp. 316-340.

- Bernard, V.L. (1989), "Capital market research in accounting during the 1980s: a critical review", paper presented at the Paper Prepared in Honor of the University of Illinois Accountancy PhD Program Golden Jubilee Symposium, University of Illinois, Urbana and Champaign, IL.
- Chandler, R.A. (1997), "Conflict, compromise and conquest in setting auditing standards: the case of the small company qualification", *Critical Perspectives on Accounting*, Vol. 8 No. 1, pp. 411-429.
- Gul, F.A. and Tsui, J.S.L. (1997), "A test of the free cash flow and debt monitoring hypotheses: evidence from audit pricing", *Journal of Accounting and Economics*, Vol. 24 No. 2, pp. 219-237.
- Haskins, M. and Williams, D. (1990), "A contingent model of intra-big eight auditor changes", Auditing: A Journal of Practice and Theory, Vol. 9 No. 3, pp. 55-74.
- Hayn, C. (1995), "The information content of losses", Journal of Accounting and Economics, Vol. 20 No. 2, pp. 125-153.
- Kim, I. (2012), "Market reactions to qualified audit reports: research approaches", Accounting Research Journal, Vol. 25 No. 1, pp. 8-24.
- Knechel, R., Naiker, V. and Pacheco, G. (2007), "Does industry specialization matter? Evidence from market reaction to auditor switches", *Auditing: A Journal of Practice and Theory*, Vol. 26 No. 1, pp. 19-45.
- Lennox, C. (2000), "Do companies successfully engage in opinion-shopping? Evidence from the UK", *Journal of Accounting and Economics*, Vol. 29 No. 3, pp. 321-337.
- Lev, B. (1989), "On the usefulness of earnings and earnings research: lessons and directions from two decades of empirical research", *Journal of Accounting Research*, Supplement, pp. 153-192.

#### Appendix

information content Variable definition Variable name Variable measurement Continuous variables R Stock returns Change in stock market price, plus the annual dividends for the present fiscal 167year divided by the beginning of the period stock price. The annual period for the estimation of returns is extending for a period of 9 months prior to the fiscal year end, and 3 months after that EARN Deflated earnings Annual earnings per share divided by beginning of the period stock price  $\Delta EARN$ Deflated earnings changes Change in annual earnings per share divided by the beginning of the period stock price Dichotomous dummy variables UNQOP It states that the financial statements Auditor's unqualified opinion present fairly, in all material respects, the financial position, results of operations and cash flows of the entity in conformity with generally accepted accounting principles. The dichotomous variable created equals to 1 if Auditor's Report type falls in this category and 0 otherwise Auditor's unqualified opinion with UNQEXP Certain circumstances, while not affecting explanatory paragraph the auditor's unqualified opinion on the financial statements, may require that the auditor add an explanatory paragraph (or other explanatory language) to the report. The dichotomous variable equals to 1 if Auditor's Report type falls in this category and 0 otherwise Auditor's qualified opinion QOPA qualified opinion states that, except for the effects of the matter(s) to which the qualification relates, the financial statements present fairly, in all material respects, the financial position, results of operations and cash flows of the entity in conformity with generally accepted accounting principles. The dichotomous variable equals to 1 if Auditor's Report type falls in this category and 0 otherwise Audit firm size AUDITOR Dichotomous variable, equal to 1 if auditing firm performing the audit belonged to either, Big 8, Big 6, Big 5 or Big 4 during the period 1981-2011, 0 otherwise SIZE Auditee size Dichotomous variable, equal to 1 if the logarithm of market value of equity for each audited firm is smaller than the median logarithm of market value of equity (Small firm), 0 otherwise (Big firm) Table AI. Variable definitions table (continued)

Incremental

JAAR	Variable definition	Variable name	Variable measurement
17,2	Financially distressed	FD	Dichotomous variable, equal to 1 if Altman's Z-Score for each audited firm is smaller than 1.81, (financially distressed firm), 0 otherwise (non-financially distressed firm)
168	Auditor's unqualified opinion issued by big auditing firms	UNQA	Intercept variable representing the product UNQOP × AUDITOR
	Auditor's unqualified opinion with explanatory paragraph issued by big auditing firms	UNXPA	Intercept variable representing the product $UNQEXP \times AUDITOR$
	Auditor's qualified opinion issued by big auditing firms	QOPA	Intercept variable representing the product $QOP \times AUDITOR$
	Auditor's unqualified opinion received by small-sized auditees	UNQSZ	Intercept variable representing the product $UNQOP \times SIZE$
	Auditor's unqualified opinion with explanatory paragraph received by small- sized auditees	UNXPSZ	Intercept variable representing the product $UNQEXP \times SIZE$
	Auditor's qualified opinion received by small-sized auditees	QOPSZ	Intercept variable representing the product $QOP \times SIZE$
	Auditor's unqualified opinion received by financially distressed auditees	UNQFD	Intercept variable representing the product UNQOP×FD
	Auditor's unqualified opinion with explanatory paragraph received by financially distressed auditees	UNXPFD	Intercept variable representing the product $UNQEXP \times FD$
	Auditor's qualified opinion received by financially distressed auditees	QOPFD	Intercept variable representing the product $QOP \times FD$
	Control variables		
	Growth proxy	GROWTH	Firm's market value of equity for year $t$ divided by firm's book value of equity for year $t$
	Leverage proxy	LEV	Firm's long-term debt for year $t$ divided by firm's total assets for year $t$
	Growth and auditor's unqualified opinion cross-term proxy	UNQG	Slope variable representing the product $UNQOP \times GROWTH$
	Growth and auditor's unqualified opinion with explanatory paragraph cross-term proxy	UNXPG	Slope variable representing the product $UNQEXP \times GROWTH$
	Growth and auditor's qualified opinion cross-term proxy	QOPG	Slope variable representing the product $QOP \times GROWTH$
	Leverage and auditor's unqualified opinion cross-term proxy	UNQL	Slope variable representing the product $UNQOP \times LEV$
	Leverage and auditor's unqualified opinion with explanatory paragraph cross-term proxy	UNXPL	Slope variable representing the product $UNQEXP \times LEV$
	Leverage and auditor's qualified opinion cross-term proxy	QOPL	Slope variable representing the product $QOP \times LEV$
Table AI.	<b>Notes:</b> All the data were available by Con AMEX and NASDAQ.UNQEXP variable v	mpustat North . was available af	America and refer to firms listed in NYSE, iter 1988

<b>About the authors</b> Panayiotis Tahinakis is based at the Department of Accounting and Finance, School of Business Administration, University of Macedonia, Thessaloniki, Greece. Panayiotis Tahinakis is the corresponding author and can be contacted at: tabi67@uom gr	Incremental information content
Michalis Samarinas is based at the Department of Accounting and Finance School of	
Business Administration, University of Macedonia, Thessaloniki, Greece and Department	
of Business Administration and Economics, The University of Sheffield, International Faculty (City College).	169